Author's response to reviews

Title: Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes

Authors:

Gillian Paul Ms (gpaul@tcd.ie)
Tom O'Dowd Prof (todowd@tcd.ie)
David Whitford Prof (dwhitford@rcsi.ie)
Fergus O'Kelly Prof (feokelly@tcd.ie)
Susan M Smith Dr (susmith@tcd.ie)

Version: 2 Date: 22 May 2007

Author’s response to reviews: see over
Re: Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes (MS: 1975486050135708)

Dear Dr Le Good,

Thank you for considering this paper for publication in BMC Health Services Research. We welcomed the reviewer’s considered and detailed comments. We have rewritten the paper based on these comments and suggestions. We have highlighted our responses to each comment or section of comments and the exact changes made in the revised manuscript. These amendments are detailed on the ‘Cover letter’ webpage.

We hope the amendments are to the reviewer’s satisfaction. Please contact us if you have any queries.

Many thanks,

__________________
Gillian Paul
Susan M Paul
Fergus O’Kelly
David Whitford
Tom O’Dowd